Author's response to reviews

Title: Meta-analysis of radiofrequency ablation versus hepatic resection for small hepatocellular carcinoma

Authors:

Yanming Zhou (zhouyms@yahoo.com.cn)
Yanfang Zhao (zhyf715@126.com)
Bin Li (Binl1962@yahoo.com.cn)
Donghui Xu (thk579@sina.com)
Zhengfeng Yin (yinzfk@yahoo.com.cn)
Feng Xie (hunanxiefeng@yahoo.com.cn)
Jiamei Yang (yjm.1952@yahoo.com.cn)

Version: 7 Date: 8 July 2010

Author's response to reviews:

Dear editor:

Thank you for your kindness to review our manuscript (MS: 1678670284339794). Following the criticisms from you, we have revised our paper.

With best wishes

Dr. Yanming Zhou
Department of Hepato-Biliary-Pancreato-Vascular Surgery
The First affiliated Hospital of Xiamen University,
Xiamen 361003, Fujian Province
China

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].' Reference 24 is currently cited after references 25 - 29 in the manuscript.

Answer: we corrected the cited references numbers.

Please make sure you use the latest version of the main file as we have made several formatting changes and using a previous unformatted version will delay publication.

Answer: yes

Typography: Please take this opportunity to check your manuscript for any
typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.